Germline Mutations in *MAP3K6* Are Associated with Familial Gastric Cancer



Daniel Gaston^{1,9}, Samantha Hansford^{2,9}, Carla Oliveira³, Mathew Nightingale¹, Hugo Pinheiro³, Christine Macgillivray⁴, Pardeep Kaurah², Andrea L. Rideout⁵, Patricia Steele⁵, Gabriela Soares⁶, Weei-Yuarn Huang⁷, Scott Whitehouse¹, Sarah Blowers⁸, Marissa A. LeBlanc¹, Haiyan Jiang⁹, Wenda Greer¹, Mark E. Samuels^{1,10}, Andrew Orr^{1,4}, Conrad V. Fernandez¹¹, Jacek Majewski¹², Mark Ludman^{5,13}, Sarah Dyack^{5,11}, Lynette S. Penney^{5,11}, Christopher R. McMaster¹⁴, David Huntsman², Karen Bedard^{1*}

1 Department of Pathology, Dalhousie University, Halifax, Nova Scotia, Canada, 2 Department of Pathology and Laboratory Medicine, University of British Columbia, Vancouver, British Columbia, Canada, 3 Expression Regulation in Cancer Group, IPATIMUP, Institute of Molecular Pathology and Immunology of the University of Porto & Medical Faculty of the University of Porto, Porto, Portugal, 4 Department of Ophthalmology and Visual Sciences, Dalhousie University, Halifax, Nova Scotia, Canada, 5 Medical Genetics, IWK Health Centre, Halifax, Nova Scotia, Canada, 6 Center of Medical Genetics Jacinto de Magalhães, Porto Hospital Center, Porto, Portugal, 7 Division of Anatomical Pathology, Department of Pathology, Queen Elizabeth II Health Science Center and Dalhousie University, Halifax, Nova Scotia, Canada, 8 Queen's Family Health Team, Kingston, Ontario, Canada, 9 Department of Biostatistics, Princess Margaret Cancer Centre, Toronto, Ontario, Canada, 10 Centre de Recherche du CHU Ste-Justine and Department of Medicine, University of Montreal, Montreal, Quebec, Canada, 11 Department of Pediatrics, Dalhousie University, Halifax, Nova Scotia, Canada, 12 Department of Human Genetics, McGill University, Montreal, Québec, Canada, 13 Oncogenetics Service, Institute of Medical Genetics, Meir Medical Center, Kfar Saba, Israel, 14 Department of Pharmacology, Dalhousie University, Halifax, Nova Scotia, Canada

Abstract

Gastric cancer is among the leading causes of cancer-related deaths worldwide. While heritable forms of gastric cancer are relatively rare, identifying the genes responsible for such cases can inform diagnosis and treatment for both hereditary and sporadic cases of gastric cancer. Mutations in the E-cadherin gene, *CDH1*, account for 40% of the most common form of familial gastric cancer (FGC), hereditary diffuse gastric cancer (HDGC). The genes responsible for the remaining forms of FGC are currently unknown. Here we examined a large family from Maritime Canada with FGC without *CDH1* mutations, and identified a germline coding variant (p.P946L) in mitogen-activated protein kinase kinase kinase 6 (*MAP3K6*). Based on conservation, predicted pathogenicity and a known role of the gene in cancer predisposition, *MAP3K6* was considered a strong candidate and was investigated further. Screening of an additional 115 unrelated individuals with non-*CDH1* FGC identified the p.P946L *MAP3K6* variant, as well as four additional coding variants in *MAP3K6* (p.F849Sfs*142, p.P958T, p.D200Y and p.V207G). A somatic second-hit variant (p.H506Y) was present in DNA obtained from one of the tumor specimens, and evidence of DNA hypermethylation within the *MAP3K6* gene was observed in DNA from the tumor of another affected individual. These findings, together with previous evidence from mouse models that *MAP3K6* acts as a tumor suppressor, and studies showing the presence of somatic mutations in *MAP3K6* in non-hereditary gastric cancers and gastric cancer cell lines, point towards *MAP3K6* variants as a predisposing factor for FGC.

Citation: Gaston D, Hansford S, Oliveira C, Nightingale M, Pinheiro H, et al. (2014) Germline Mutations in MAP3K6 Are Associated with Familial Gastric Cancer. PLoS Genet 10(10): e1004669. doi:10.1371/journal.pgen.1004669

Editor: Marshall S. Horwitz, University of Washington, United States of America

Received March 3, 2014; Accepted August 14, 2014; Published October 23, 2014

Copyright: © 2014 Gaston et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The following agencies provided funding for this project: Genome Canada, Genome Atlantic, Nova Scotia Health Research Foundation, Nova Scotia Research and Innovation Trust, Dalhousie Faculty of Medicine, Dalhousie Department of Ophthalmology, Health Canada, The Centre for Drug Research and Development, Capital District Health Authority, IWK Health Centre Foundation, Capital Health Research Fund, and The COMPETE/FEDER Portuguese Foundation for Science and Technology (FCT), Projects Ref. FCT PTDC/SAU-GMG/110785/2009 and Post-doc grant SFRH/BPD/79499/2011 to HP "financiados no âmbito do Programa Operacional Temático Factores de Competitividade (COMPETE) e comparticipado pelo fundo Comunitário Europeu FEDER." MES is supported by the CHU Ste-Justine Centre de Recherche. The authors would like to acknowledge the contribution of: the Genome Quebec High Throughput Sequencing Platform; and Sónia Sousa and José Carlos Machado from the IPATIMUP Diagnostics Unit, Porto, Portugal. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* Email: kbedard@dal.com

• These authors contributed equally to this work.

Introduction

Gastric cancer is the second leading cause of cancer-related death worldwide with 738,000 deaths per year [1]. Primary treatment consists of surgical resection of the tumor and may be followed by chemotherapy and/or radiotherapy. The 5-year survival rates after surgical resection are high if the disease is detected early (71% for stage 1A), however, they drop off quickly when the diagnosis is made at later stages (46% stage IIA, 20% stage IIIA, 4% stage IV) (National Cancer Institute's SEER database, October 2013). Unfortunately, because early symptoms of gastric cancer closely resemble other diseases, detection often does not occur until advanced stages have already been reached [2].

Author Summary

The underlying genetic mutations involved in 60% of inherited gastric cancer cases remain unknown. Here we present a large, extended pedigree with familial gastric cancer and an association in part of the family with a mutation in *MAP3K6*. The conservation, predicted pathogenicity of the variant, tissue distribution, and known function of *MAP3K6* made this a strong candidate that warranted further investigation. Examination of an additional 115 unrelated probands identified additional mutations in *MAP3K6*, including a truncating mutation.

Classically, gastric cancer has been divided into two types: intestinal and diffuse [3]. The intestinal form occurs spontaneously and is most often found in elderly individuals, while the diffuse form often occurs in younger individuals and can be associated with a family history of gastric cancer. Populations with higher prevalence of chronic *Helicobacter pylori* infection tend to have higher gastric cancer burdens [4]. The majority of gastric cancers (90%) are sporadic, but approximately 10% show familial clustering [5]. Only 1% to 3% are caused by a hereditary syndrome, as opposed to environmental factors such as shared dietary practices [5]. The most well established familial form of gastric cancer is hereditary diffuse gastric cancer (HDGC [MIM 137215]), where approximately 40% of cases are attributed to germline mutations in the E-cadherin encoding gene, *CDH1* [6–9].

We ascertained a large family from Maritime Canada with a history of Familial Gastric Cancer (FGC) displaying an apparent autosomal dominant pattern of inheritance, but bearing no variants in the coding region of the *CDH1* gene. While the family displays many features typical of HDGC, there was diversity in the clinical presentation within the family as well as an advanced age of onset, therefore we have opted to simply refer to the condition as FGC over the more stringently defined HDGC. Genomic mapping of shared inherited regions among affected family members, followed by whole-exome sequencing, led to the identification of a germline single nucleotide variant (SNV) in mitogen-activated protein kinase kinase kinase 6 (MAP3K6, ASK2, MAPKKK6, MEKK6, ENSG00000142733), a gene encoding a member of the serine/threonine protein kinase family. Several in silico methods predicted the SNV in MAP3K6 to be damaging to the protein, and previous studies with MAP3K6 deficient mice [10], as well as the occurrence of mutations in this gene in both primary gastric cancer tumors and gastric cancer cell lines [11], were consistent with mutations in the MAP3K6 gene being the causative mutation. Sequencing of DNA isolated directly from a fixed tumor specimen of one individual demonstrated the presence of a *de novo* second-hit variant in MAP3K6. Screening of an additional 115 unrelated FGC samples, also negative for CDH1 mutations, revealed five individuals with four additional SNVs in MAP3K6 that were also predicted to be pathogenic, as well as an unrelated individual with the SNV identified in the family from Maritime Canada. The age of onset varied among MAP3K6 SNV carriers in the five families, and one had not developed cancer even at late stage of life, suggesting incomplete penetrance. This is the first report of a heritable cancer resulting from SNVs in MAP3K6.

Results

Clinical and Pathological Assessment

We ascertained a large Maritime Canadian family of European descent in the course of routine clinical assessment in the Hereditary Cancer Clinic as part of Maritime Medical Genetics Service at the IWK Health Centre in Halifax, Nova Scotia, Canada (Figure 1). Saliva, blood, or formalin-fixed paraffinembedded (FFPE) samples were obtained from 6 family members with gastric cancer, as well as 27 unaffected relatives, and one married-in individual. No consanguinity was suspected in this pedigree.

The proband, affected individual 1884, was diagnosed with metastatic gastric carcinoma and underwent a total gastrectomy at age 51. Pathological examination revealed a poorly differentiated adenocarcinoma arising in the antrum of the stomach in a background of intestinal metaplasia and chronic gastritis. The tumor was composed of a sheet of signet ring cells (Figure 2C and D). The carcinoma penetrated through the entire thickness of the muscularis propria involving the serosal layer. No evidence of H. *pylori* was seen.

The proband's maternal aunt, individual 1826, was diagnosed with gastric carcinoma at age 80 and underwent a partial gastrectomy. Pathological examination revealed a moderately to poorly differentiated adenocarcinoma invading into muscularis propria (Figure 2A). The tumor was predominantly composed of cohesive nests of neoplastic cells with occasional glandular formation. Tumor cells with signet ring cell forms were seen in solid areas (Figure 2B). The gastric mucosa adjacent to the tumor showed focal intestinal metaplasia without evidence of *H. pylori*.

At age 76 a stomach biopsy of another maternal aunt to the proband, individual 1841, was reported to have a moderately differentiated adenocarcinoma with glandular formation. *H. pylori* was identified in the background gastric mucosa. A small biopsy section was available for re-examination. Although this sample was too small for a complete classification, it showed cohesive nests of neoplastic cells with small foci of glandular formation in keeping with a poorly differentiated adenocarcinoma. There were some tumor cells in the sample showing clear cytoplasm, but these could not be definitively classified as signet ring cells.

A stomach biopsy at age 82 of patient 1844, the proband's mother, showed a poorly differentiated adenocarcinoma with signet ring features. The background mucosa showed evidence of *H. pylori* and patchy intestinal metaplasia.

Patient 1845, a first-cousin-once-removed to the proband, was diagnosed at age 59 with an undifferentiated carcinoma without signet ring features. The tumor was associated with dense lymphoid infiltrate and was best classified as lymphoepithelial carcinoma. *H. pylori* was not seen in the adjacent normal mucosa. The tumor was positive for two intronic variants in *CDH1* both of which are expected to be benign (NM_004360.3:c.688-83G>A and c.2439+52G>A).

Patient 2447, a third cousin, was diagnosed at age 44 with a poorly differentiated adenocarcinoma without signet ring cell features. The adjacent gastric mucosa showed extensive intestinal metaplasia. There was no evidence of *H. pylori*.

Following screening of a panel of 115 probands with non-*CDH1* familial gastric cancer, an unrelated family from Portugal was added to our study (Figure 3A). Individual II-6 was diagnosed with gastric cancer at age 62, having poorly differentiated adenocarcinoma of the stomach and the presence of signet ring cells. Immunohistochemistry analysis showed positive membranous staining of E-cadherin in neoplastic cells (Figure 3C), including signet ring cells (Figure 3B). The related individuals I-4, II-1, and II-7 were diagnosed with gastric cancer (histology details unknown) at ages 53, 62 and 52 respectively. All four individuals in the Portuguese pedigree died from the disease within 5 years of diagnosis in this family. In the Maritime Canadian





Figure 1. Pedigree of the Maritime Canadian family. Clinically affected individuals are indicated with shaded symbols. Individuals for whom DNA samples were collected are indicated by a number. Individuals shaded within 2 quarter sections were affected but were negative for the MAP3K6 mutation. Individuals shaded on one half had another, non-gastric, cancer. Generations I–VI are indicated. doi:10.1371/journal.pgen.1004669.g001



Figure 2. Histology of two gastric cancer patients from the Maritime Canadian family. A) Patient 1826. Section shows a tumor composed of solid and glandular components (H&E, $40 \times$). **B**) The tumor cells show signet ring cell morphology in solid areas (H&E, $200 \times$). **C**) Patient 1884. Section shows sheet of tumor cells infiltrated lamina propria and submucosa (H&E, $40 \times$). **D**) The tumor entirely consists of signet ring cells (H&E $200 \times$).

doi:10.1371/journal.pgen.1004669.g002



Figure 3. A) **Pedigree of the Portuguese Familial Gastric Cancer family.** Affected individuals are shaded in black with the sequenced proband indicated with a triangle. Deceased individuals are marked with a strike-through. Generations I–III are indicated. **B**) Tumor cells showing signet ring cell morphology (H&E, 200×). **C**) Tumor cells retaining E-cadherin protein expression (IHC analysis performed with the rabbit anti-E-cadherin Antibody (24E10 Cell Signaling, MA, USA), according to manufacturer's instructions, 200×). doi:10.1371/journal.pgen.1004669.q003

family, 1884, 1844 and 1841 died from the disease within one year of diagnosis.

The gastric cancer described for patients 1845 and 2447 had no signet rings observed and was diagnosed at an earlier average age (52 versus 72, although the proband was diagnosed at age 51). Based on differences in histology, particularly the lack of signet ring cells in 1845 and 2447 compared to the other affected individuals, it is possible that the disease in these two individuals represents a distinct condition. Alternatively it is possible that the family is displaying a more complex phenotypic pattern being driven by two (or more) genes.

Molecular Mapping and Exclusion of Known and Candidate Genes

Although 30-40% of HDGC cases are attributable to mutations in *CDH1*, no mutations in protein-coding exons of *CDH1* were found in affected individuals from the Maritime Canadian family. To identify the pathogenic loci in this family, high density SNPgenotyping using Illumina arrays was performed on five affected individuals: the proband's mother (1844), two affected maternal aunts (1826 and 1841), and two distant cousins (1845 and 2447) as well as several related individuals with no reported incidence of cancer whose affection status was treated as unknown (1907, 1924, 1821, and 1822). For all individuals except 1845 genotyping data was available at 2.5 million markers. Individual 1845 had previously been genotyped at a density of 660K, and was not able to be re-genotyped at the higher density. No DNA suitable for SNP genotyping was obtained from the FFPE sample of the proband (1884). Using these data, we performed both nonparametric and parametric linkage analysis using Merlin [12]. Given the late age of onset in many affected family members, the penetrance in the Maritime pedigree is unknown. In order to be conservative in identifying genomic regions of interest, two dominant penetrance models (50% and 99% penetrance) using affected individuals 1826, 1841, 1844, and 2447 (and individuals 1907, 1924, 1821, and 1822 with unknown affection status) were used. Genomic regions identified under parametric linkage analysis were generally consistent with one another regardless of the penetrance parameter chosen (Table 1). The analysis was repeated with individual 2447 treated as unknown to analyze just the reduced pedigree where 2447 and 1845 were treated as potential phenocopies (Table 2). This resulted in lower overall LOD scores for all regions identified, as well as more and larger regions on average, encompassing a larger portion of the genome.

We also performed non-parametric linkage (NPL), a method with fewer underlying assumptions about the underlying inheritance model, using affected individuals 1826, 1841, 1844, and 2447 (pedigree-wide) or with the removal of 2447 as a potential phenocopy by specifying them to be of unknown status (subpedigree). Pedigree-wide genomic intervals were mostly consistent with those identified using the two parametric models (Table 3). Exclusion of 2447 resulted in a lower overall maximum score (1.204), which was found on several intervals throughout the genome (Table 3).

Genomic intervals identified in this manner were used for filtering the exome sequencing data to identify potential causative mutations. To be broad in identifying possible causative mutations, in both the pedigree-wide and sub-pedigree case the intervals from the respective parametric and non-parametric analyses were combined. For the pedigree-wide analyses this was the union of intervals described in Table 1 along with the appropriate intervals in Table 3 (including 2447) and for the subpedigree analysis the union of intervals found in Table 2 with appropriate intervals in Table 3 (excluding 2447).

Whole Exome Sequencing

We next performed whole-exome sequencing on two of the affected maternal aunts to the proband (1826 and 1841) and the

Table 1. Regions with LOD>1 from pedigree-wide parametric linkage analyses using Merlin.

Penetrar	nce: 99%				
Chr	Max LOD	Start SNP	Start bp	End SNP	End bp
1	1.0678	rs1409162	71,368,758	rs871664	94,609,478
3	1.7039	rs938087	194,291,805	rs2686097*	197,304,279
7	1.6951	rs4722077	22,011,292	rs10281171	36,807,381
10	1.5943	rs7923552	21,687,123	rs10508881	44,541,565
15	1.5321	rs17626899	94,215,925	rs11073574	98,306,216
16	1.1006	rs9302859	7,989,791	rs4785367	49,956,194
17	1.6395	rs1269480	49,815,361	rs9897212	60,190,630
20	1.6937	rs4501008	51,433,681	rs6027887	59,449,041
Penetrar	nce: 50%				
Chr	Max LOD	Start SNP	Start bp	End SNP	End bp
3	1.4762	rs938087	194,291,805	rs2686097*	197,304,279
7	1.4597	rs4722077	22,011,292	rs10281171	36,807,381
10	1.3718	rs9645525	18,620,950	rs2153332	45,731,944
15	1.3014	rs17626899	94,215,925	rs11073574	98,306,216
16	1.471	rs8055473	9,798,539	rs4785367	49,956,194
17	1.4055	rs1269480	49,815,361	rs9897212	60,190,630
20	1.4584	rs4501008	51,433,681	rs6027887	59,449,041

Genomic intervals and associated LOD scores are shown under dominant models with 50% or 99% penetrance. Regions are defined by their 1-LOD support interval. Base pair positions are from hg19. An asterisk indicates the SNP was the first or last analyzed marker on the chromosome.

doi:10.1371/journal.pgen.1004669.t001

affected third cousin (2447). We prioritized and filtered variants based on their frequency among European-descent populations (< 2% and a stricter filter at <1%) from the 1000 Genomes and Exome Sequencing Project datasets as well as other exomes sequenced at the same sequencing provider, location within a genomic region of interest, and the functional consequence of the mutation (altering the protein coding sequence or splice site of at least one protein coding transcript). Variants of interest were then sequenced by Sanger sequencing in other affected individuals. Variant filtration based on genomic intervals was performed separately for each hypothesis (whole-pedigree and reducedpedigree) (Tables S1 and S2). In addition to the identification and filtration of genetic variants, we assessed the sequencing depth of coverage of exons (defined by the Consensus CDS set) within genomic regions of interest and across individual exome sequencing results. Further, we searched for potentially shared variants that were "masked" by coverage issues. For all variants observed in one or more exomes, if no variant was observed in the remaining exome(s), we evaluated whether that was due to low coverage or coverage gaps within the exon. For variants where this was true we filtered using standard criteria (as above). Using these filtering criteria, several variants with low MAF and potentially having an effect at the protein-coding level were observed in the pedigree-wide genomic regions of interest that had been identified by parametric linkage analysis; however, none were present in all affected individuals. Further no "masked" candidate variants were identified by the same criteria.

We considered the possibility that individuals 1845 and 2447 have a distinct clinical condition, and examined the variants shared among the proband and immediate family. Using the same filtering criteria as above, but using only the exomes from individuals 1826 and 1841, a total of 127 variants were identified. Stricter filtering for rare variants (MAF <1%) reduced this to number to 85 (Table S2). A subset of these variants, based on a combination of factors (mutations in COSMIC [13], predicted effect of the mutation, conservation of the encoded amino acid, literature review, known expression patterns in normal tissues and tumors, disease phenotypes associated with the gene) were sequenced for follow-up in the proband and their mother. A variant in MAP3K6 (Chr1, NM_004672) was of particular interest. A mutation identified in MAP3K6 (c.[2837C>T];[=], p.P946L) was considered a strong candidate based on the known associations of other MAP kinases with cancer, and several publications elucidating a role for MAP3K6 in tumorigenesis [10,11,14,15]. This variant has been reported previously (rs141787524) with a minor allele frequency of 0.7% in the 1000 Genomes Project (European descent group) and a frequency of 0.4% in the European-American population (Exome Variant Server (NHLBI GO Exome Sequencing Project (ESP): http://evs. gs.washington.edu/EVS [Accessed October, 2013]). It was seen as a heterozygous variant in 11 (of 1532) other exomes sequenced at the Genome Quebec Innovation Centre, corresponding to a MAF of 0.36%.

This SNV was present in four affected individuals in the Maritime family (1884, 1826, 1844, 1841), of which three clearly showed the presence of signet ring cells. Only a small punch biopsy was available for the maternal aunt, 1841, therefore we were not able to definitively confirm the presence or absence of signet ring cells. The MAP3K6 SNV was also present in five of the 27 currently unaffected relatives sampled, and it was not present in the married-in relative. One of the carriers was homozygous for the SNV and was over 80 years old with no reported cancer.

Table 2. Regions with LOD>0.5 from parametric linkage analyses using Merlin when individual 2447 is treated as unknown (sub-pedigree).

Penetrar	nce: 99%				
Chr	Max LOD	Start SNP	Start bp	End SNP	End bp
1	0.5692	rs1702312	25,364,652	rs2039943	102,297,845
2	0.5692	rs6718709	5,340,502	rs7562738	104,060,568
3	0.5692	rs2363970	190,873,984	rs2686097*	197,304,279
5	0.5692	rs4957023*	336,952	rs1604563	3,718,709
5	0.5692	rs1437118	94,301,042	rs889056	157,039,737
6	0.5692	rs6454657	88,571,309	rs3736746	144,259,554
7	0.569	rs12530679	106,632,113	rs850398	145,130,880
8	0.5692	rs12156116	138,551,517	rs750472*	145,701,453
9	0.5692	rs1454637	2,596,569	rs912670	115,673,030
10	0.5692	rs7906313*	766,105	rs10762712	54,112,284
12	0.5692	rs10846635	124,728,014	rs7296889*	133,054,656
13	0.5692	rs7992848	32,800,816	rs9571607	67,204,131
14	0.5692	rs4497606*	20,669,142	rs225904	30,454,892
16	0.5692	rs9302859	7,989,791	rs8046479	83,227,014
17	0.5692	rs7215862	14,282,778	rs9897212	60,190,630
20	0.5691	rs4813237	16,635,038	rs6027887	59,449,041
22	0.5586	rs2337542	47,758,290	rs2688098	49,682,956
23	0.5692	rs6567564*	3,510,277	rs5945157*	152,901,900
Penetrar	nce: 50%				
Chr	Max LOD	Start SNP	Start bp	End SNP	End bp
1	0.5163	rs11249206	25,277,982	rs2208963	104,529,736
2	0.5163	rs6718709	5,340,502	rs7562738	104,060,568
3	0.5163	rs2363970	190,873,984	rs2686097*	197,304,279
5	0.5163	rs4957023*	336,952	rs580641	4,041,154
5	0.5163	rs1437118	94,301,042	rs17608675	157,340,772
6	0.512	rs6454657	88,571,309	rs3736746	144,259,554
7	0.5162	rs12530679	106,632,113	rs850398	145,130,880
8	0.5163	rs12156116	138,551,517	rs750472*	145,701,453
9	0.5163	rs1454637	2,596,569	rs1570756	115,775,168
10	0.5163	rs7906313*	766,105	rs10762712	54,112,284
12	0.5163	rs4765078	124,646,827	rs7296889*	133,054,656
13	0.5163	rs9600625	31,470,643	rs9571607	67,204,131
14	0.5163	rs4497606*	20,669,142	rs225904	30,454,892
16	0.5163	rs9302859	7,989,791	rs8046479	83,227,014
17	0.5163	rs7215862	14,282,778	rs9897212	60,190,630
20	0.5162	rs4813237	16,635,038	rs6027887	59,449,041
22	0.5063	rs2337542	47,758,290	rs2688098	49,682,956
23	0.5163	rs6567564*	3,510,277	rs5945157*	152,901,900

Genomic intervals and associated LOD scores are shown under dominant models with 50% or 99% penetrance. Regions are defined by their 1-LOD support interval. Base pair positions are from hg19. An asterisk indicates the SNP was the first or last analyzed marker on the chromosome. doi:10.1371/journal.pgen.1004669.t002

Although no consanguinity was reported in the family, and no evidence of copy number variation was observed in the SNP genotype data, this individual was also homozygous for a 10 Mb region encompassing the locus. The remaining carriers ranged in age from 33 to 51, and as the age of onset of the cancer was generally later, their status was considered "unknown". Both

individuals, 1845 and 2447, with the phenotypically distinct gastric cancer were negative for the *MAP3K6* SNV.

Somatic Variants within the Tumor

We next used the DNA isolated from a tumorous section of the FFPE sample of the affected MAP3K6 SNV carrier 1884 (the

Table 3. Genomic intervals with NPL>2 or >1 by non-parametric linkage analysis using the S_{all} scoring function, under the exponential model of Kong and Cox. Results including 2447 (NPL>2) or excluding 2447 (NPL>1) are shown.

Including	a 2447						
Chr	MaxNPL	p-value	start SNP	start bp	end SNP	end bp	
1	2 001	0.0012	rs4607942	76 750 428	rs472908	94 487 354	
3	2 709	0.0002064	rs938087	194 291 805	rs2686097*	197 304 279	
10	2 593	0.0002743	rs7923552	21 687 123	rs12781751	31 872 826	
10	2 482	0.0003609	rs7911097	36.651.214	rs10508881	44.541.565	
16	2.703	0.0002094	rs8055473	9.798.539	rs4785367	49.956.194	
Evolution	- 2447			-,		,	
Excludin	ig 2447						
Chr	MaxNPL	p-value	start SNP	start bp	end SNP	end bp	
1	1.204	0.009266	rs3767879	26,070,909	rs545247	29,174,044	
1	1.204	0.009266	rs545247	29,174,044	rs9326132	43,471,182	
1	1.204	0.009266	rs11206577	55,860,561	rs11578884	70,282,875	
1	1.204	0.009266	rs11578884	70,282,875	rs472908	94,487,354	
2	1.204	0.009266	rs11675383	20,882,949	rs2374292	41,849,795	
3	1.204	0.009266	rs938087	194,291,805	rs2686097*	197,304,279	
5	1.204	0.009266	rs4957023*	336,952	rs2922062	2,618,844	
5	1.204	0.009266	rs10065787	131,436,486	rs853160	142,559,092	
6	1.204	0.009266	rs6901395	89,432,362	rs11153525	115,231,965	
6	1.195	0.009491	rs11153525	115,231,965	rs4382328	123,022,496	
7	1.204	0.009266	rs7777931	135,016,454	rs12703526	143,107,588	
8	1.204	0.009266	rs7007907	139,914,637	rs2233230	140,943,503	
9	1.204	0.009266	rs7025491	37,387,177	rs11138454	82,623,065	
9	1.155	0.01055	rs11138454	82,623,065	rs7849075	86,149,610	
10	1.204	0.009266	rs7906313*	766,105	rs10905707	10,321,874	
10	1.204	0.009266	rs2446580	14,005,536	rs1774221	30,254,563	
12	1.204	0.009266	rs7135277	128,737,601	rs7296889*	133,054,656	
13	1.204	0.009266	rs2281810	43,461,643	rs8002829	49,951,142	
14	1.204	0.009266	rs4497606*	20,669,142	rs761874	23,596,320	
16	1.204	0.009266	rs150583	21,712,743	rs4620978	62,192,704	
17	1.204	0.009266	rs3764866	26,800,204	rs3916033	44,889,703	
20	1.204	0.009266	rs6028809	38,663,214	rs4142387	42,017,672	
22	1.182	0.009834	rs133651	48,717,026	rs9617453	49,034,053	
23	1.204	0.009266	rs6567564*	3,510,277	rs2188385	9,923,646	

An asterisk indicates the SNP was either the first or last marker on the chromosome. Base pair positions are from hg19. doi:10.1371/journal.pgen.1004669.t003

proband) to screen for additional somatic SNVs or loss of heterozygosity (LOH) within the tumor itself. In addition to the p.P946L variant, we identified a novel SNV in the MAP3K6 gene at position c.[1516C>T] leading to an amino acid change p.H506Y (Table 4), and were able to infer that the SNV was somatically acquired based on sequence data from the spouse and children.

Verification in Unrelated FGC Cases

We screened DNA samples from an additional 115 unrelated FGC individuals using a multiplexed targeted next generation sequencing assay. Samples were from unrelated families that met international gastric cancer linkage consortium (IGCLC) criteria for hereditary diffuse gastric cancer (106), but had previously tested negative for mutation of the *CDH1* locus, or familial intestinal gastric cancer. Within this cohort, we identified five

additional heterozygous SNVs in the MAP3K6 gene (Table 4): a truncating SNV (c.[2544delC], p.F849Sfs*142), three missense SNVs (c.[2872C>A], p.P958T; c.[598G>T], p.D200Y; and c.[620T>G], p.V207G), and another individual with the p.P946L (c.[2837C>T]) variant, previously found in the Maritime family (Figure 4). Mutations in MAP3K6 were only discovered in the individuals meeting the criteria for diffuse gastric cancer (106). The individual in this novel patient cohort carrying the p.P946L variant is thought to be unrelated to the Maritime family.

The truncating SNV was observed in a Portuguese individual with a family history of gastric cancer (Figure 3A). While this SNV has a dbSNP identifier (rs34008139), no population frequency has been associated with it from either the 1000 Genomes or Exome Variant Server projects. This SNV has also been reported in the COSMIC database [13] (somatic/germline status not specified) in

Table 4. Summary information for each of the germline and somatic mutations found in MAP3K6.

Position (hg19)	Amino Acid Change	HGVS (NM_004672)	dbSNP ID	MAF (1000G/EVS) (%)
Chr1: 27690792	D200Y	c.598G>T	rs41291098	0.4/0.4
Chr1: 27690770	V207G	c.620T>G	rs182712391	-/0.01
Chr1: 27688258	H506Y ^a	c.1516C>T	-	-/-
Chr1: 27684750	P946L	c.2837C>T	rs141787524	0.7/0.4
Chr1: 27684715	P958T	c.2872C>A	rs75893867	-/-
Chr1: 27685238-27685239	p.F849Sfs*142	c.2544delC	rs34008139	-/-

^aThe H506Y mutation is a somatic second-hit observed in FFPE tumor tissue from patient 1884.

doi:10.1371/journal.pgen.1004669.t004

a carcinoma sample of the large intestine. Histological examination of the proband's tumor showed a poorly differentiated gastric cancer with signet ring cells, retaining E-cadherin protein expression at the cell membrane (Figure 3B and 3C).

The p.D200Y variant was found in probands from two unrelated families, and has been observed within the 1000 genomes cohort (rs41291098), but is also rare with a minor allele frequency (MAF) of 0.4% in both the 1000 Genomes and Exome Sequencing Project European descent groups. The p.P958T variant (rs75893867) has no MAF reported in either the 1000 Genomes or Exome Sequencing Project European descent datasets, and was only identified in the 1000 Genomes among the Japanese cohort at a frequency of 2.2%. However, it has been identified in COSMIC (COSM99077) as a somatic mutation from a gastric carcinoma patient. The p.V207G variant has been identified in the Exome Sequencing Project European-American dataset with a MAF of 0.01%. It was not identified among European or European descent groups within the 1000 Genomes, but was observed within other population groups at a range of minor allele frequencies.

Along with *MAP3K6*, 50 additional genes previously suggested to be involved in risk for disease of the upper gastrointestinal tract were sequenced for this cohort using a custom panel-based assay (manuscript submitted). Genes for the custom MiSeq-based screen were selected based on literature review, as well as genes of interest in collaborative projects. In the cases where *MAP3K6* missense variants were identified, no other candidate variants were found.

Second-Hit Analysis in the Tumor from the Portuguese Family

FFPE tumor tissue was available for the proband from the Portuguese family carrying the p.F849Sfs*142 germline truncating mutation. Somatic mutations were excluded in the complete coding sequence and intron-exon boundaries of MAP3K6 in the proband's tumor. Lack of LOH at the MAP3K6 gene could also be inferred in this tumor, as both wild-type and mutant alleles could be identified at the germline mutation site (Supplementary Figure S2). We therefore searched for putative alternative inactivating mechanisms. Hypermethylation of CpG islands within gene promoters and regulatory regions is a common phenomenon leading to decreased gene expression in cancer [16]. MAP3K6 regulation by promoter hypermethylation has been described for human bone marrow mesenchymal stem cell [17], although a correlation of hypermethylation and gene expression has not been established. We searched for MAP3K6 CpG islands [18] and found two CpG islands, one at the promoter region and another encompassing exon 10 and part of the downstream intron (Figure 5 and Supplementary Figure S1). The downstream CpG island (CpG island 2) is near a DNase hypersensitive site predicted to harbor promoter associated features (Ensembl regulatory feature ID ENSR00000533270, Figure 5). We bisulfite-treated DNA from: the proband's peripheral blood lymphocytes (PBLs); the proband's tumor; four different normal stomach control samples, and; seven gastric cancer cell lines. For the promoter CpG island (CpG island 1), no hypermethylation was detected using two different primer sets (Figure 5 and Suppl. Figure S1).







Figure 5. Methylation analysis of the Portuguese family. Left panel: Schematic representation of the MAP3K6 gene adapted from Ensembl genome browser (release 75). The two CpG islands analyzed are represented. CpG island 1 is mainly non-methylated for several normal tissues and cells lines represented in the scheme, while CpG island 2 displays low methylation frequency (light green) in normal tissues such as B-cells, Colon, Liver and Whole Blood, and high methylation (blue) in colon (HCT116), liver (HepG2) and blood (Jurkat) cancer cell lines. A DNase HSS predicted to harbor a promoter-associated regulatory element overlapping with CpG island 2. Right panel: For the CpG island 1, no hypermethylation was detected (white circles). For the CpG island 2, we observed complete methylation in the proband's tumor DNA (black circles) and no methylation in the PBLs' DNA. The DNA of normal gastric mucosa from controls displayed partial methylation (grey circles). All gastric cancer cell lines mimicked the full methylation observed for the tumor DNA (black circles). doi:10.1371/journal.pgen.1004669.g005

Regarding CpG island 2, we observed complete methylation for the tumor DNA and no methylation for the PBLs DNA. Interestingly, the methylation analysis at CpG island 2 in normal stomach mucosa from controls displayed a partial methylation pattern. In line with the result obtained for tumor DNA, all seven gastric cancer cell lines displayed full methylation (Figure 5 and Suppl. Figure S1).

Pathogenicity of SNVs in MAP3K6

We applied a variety of in silico methods to predict the pathogenicity of the observed missense SNVs. Although there was no full consensus across programs (Table 5), all of the SNVs were considered deleterious by at least one program, and except p.V207G and p.P958T, the other four variants described in this report were predicted to be deleterious by at least 3 of the 7 methods. In addition, the EvoD [19] consensus prediction (based on a balanced combination of the EvoD, PolyPhen2 [20], and SIFT [21] scores) reported that three of the variants (p.D200Y, p.V207G, p.H506Y) were deleterious or likely to be deleterious. The two remaining variants, p.P946L and p.P958T, were predicted to likely be neutral changes although they were evolutionarily ultra-conserved and well-conserved (according to EvoD evolutionary rate classification) respectively; while only one of the other three variants (p.D200Y) was considered wellconserved and the other two were less-conserved. As (i) a MAP3K6 truncating variant was found, (ii) there was a secondhit variant identified in the individual for whom FFPE tissue was examined, and (iii) all of the programs tested are designed to predict pathogenicity based on loss of function, it is likely that the variants described lead to either a decrease in function or dominant negative phenotype.

Discussion

Here we present the first evidence that germline mutations in MAP3K6 are linked to inherited cancer. Four individuals with gastric cancer from a Maritime Canadian family were found to carry a heterozygous variant in the MAP3K6 gene, leading to a p.P946L amino acid change. This germline variant, located on chromosome 1, was identified in two of the three exome samples and was located within a region identified by parametric and

non-parametric linkage analysis within the sub-pedigree (1845 and 2447 treated as "unknown" disease status). The significance of the MAP3K6 variant was supported by the identification of a somatic second-hit mutation in the MAP3K6 gene at p.H506Y present in DNA isolated directly from a tumorous section of a patient FFPE sample. Two individuals from the pedigree with gastric cancer, but with some phenotypic differences did not carry the mutation; however, no candidate variants were identified shared among all affected individuals within any region identified, by parametric linkage conducted pedigree-wide.

Screening of additional FGC families revealed five other MAP3K6 mutations, including a p.F849Sfs*142 germline mutation observed in the Portuguese proband, which is expected to lead to protein truncation. After excluding somatic mutations and LOH, a potential second-hit mechanism was found via hypermethylation at an intragenic CpG island near a predicted promoter-associated regulatory element (DNAse I hypersensitive site). The relevance of methylation at this MAP3K6 gene region could not be ascertained in terms of impact in gene expression, nevertheless the possibility of acting as a possible second-hit inactivation (partial or complete) mechanism is raised, due to the results obtained in normal stomach and cancer cell lines. If so, this may well represent another example of the increasingly recognized concept that DNA methylation in the gene body is not just a passive witness of gene transcription, but is actively involved in multiple gene regulation processes [22], warranting further investigation. Histopathology analysis of the individual from the Portuguese family carrying this truncating mutation featured signet-ring cells as part of the tumor phenotype, as did most individuals from the Maritime sub-pedigree (except 1841, where the signet ring status was inconclusive due to lack of sufficient material).

Although MAP3K6 mutations have not previously been identified in inherited cancer, there is a growing body of evidence that MAP3K6 has an important role in cancer pathogenesis. In mice, where MAP3K6 is normally expressed in gastric tissue and skin, the loss of MAP3K6 in homozygous knockout mice was found to increase susceptibility to induced skin cancer [10]. The mice did not develop cancer spontaneously; however, chemical induction performed in the presence of an inflammatory stimulus led to a greater number of skin tumors in the MAP3K6 deficient

wino Acid Change	PolyPhen2	PMut	PROVEAN	SIFT	EvoD	FATHMM*	MutationTaster
200Y	Probably Damaging	Neutral	Deleterious	Damaging	Neutral	Tolerated/Passenger	Disease Causing
207G	Benign	Neutral	Neutral	Tolerated	Deleterious	Tolerated/Passenger	Polymorphism
506Y ^a	Possibly Damaging	Neutral	Neutral	Damaging	Deleterious	Tolerated/Passenger	Polymorphism
946L	Probably Damaging	Pathological	Neutral	Tolerated	Neutral	Tolerated/Passenger	Disease Causing
958T	Possibly Damaging	Neutral	Neutral	Neutral	Neutral	Tolerated/Passenger	N/A
.F849Sfs*142							

from patient 1884 tissue in FFPE tumor ^aThe H506Y mutation is a somatic second-hit observed doi:10.1371/journal.pgen.1004669.t005

mice than in control animals. The number of tumors in heterozygous $(MAP3K6^{+/-})$ mice, as well as their size, was intermediate between the wild-type and knock-out mice, suggesting a role for MAP3K6 dosage in its effects [10]. The susceptibility of these mice to gastric cancer was not assessed.

MAP3K6 is a member of the c-Jun N-terminal Kinase (JNK) and p38 signaling cascades [14] that is most strongly expressed in the skin, gastrointestinal tract, and lungs [10]. MAP3K6 forms a heteromeric complex with its paralog MAP3K5, through their coiled-coil domains, preventing constitutive MAP3K6 degradation and promoting its autophosphorylation and activation [14]. In the presence of oxidative stress and reactive oxygen species (ROS) such as occurs following chemical (eg. DMBA) or UVA induction. active phospho-MAP3K6 activates the INK and p38 cascades and promotes apoptosis [10,14]. These effects have been seen in chemically induced skin-cancer models in $MAP3K6^{-/-}$ mice as well as primary keratinocyte cell lines [10].

While MAP3K6 has a limited tissue and cell-type distribution, MAP3K5 is more broadly expressed. The interaction of these two proteins appears to be tightly regulated, with a balance necessary between the pro-apoptotic and tumor suppressing roles of MAP3K6 and the pro-inflammatory/anti-apoptotic roles of MAP3K5 [10,14,15,23]. Expression of MAP3K6 is variable in many human tumors, with expression most significantly reduced in gastric cancer tumors compared to healthy gastric tissue [10], whereas MAP3K5 expression is increased [23]. Indeed $MAP3K5^{-/-}$ mice are more resistant to chemically induced gastric cancers than wild-type mice. It is clear that MAP3K5 and MAP3K6 are finely balanced to carry out both inflammatory and apoptotic roles, respectively in tissues where they are both expressed. Somatic mutations in MAP3K6 have been reported in screens of a panel of 532 kinase genes in apparent non-familial gastric cancer cases (p.S291L), as well as in two gastric cancer cell lines (N87 cells: p.R375Q, and IM95 cells: p.P958T) [11]. Somatic mutations in MAP3K6 have also been identified in other, nongastric cancers, including ovarian (p.T968I [24]) and breast cancers (p.P869T[25], p.S648L, p.Q672* [24])). Furthermore, although expression of MAP3K6 is variable in cervical and ovarian cancer, there is a decrease in MAP3K6 expression, compared to normal tissue, in 75% out of 106 oral, esophageal, gastric and colorectal cancer cell lines tested [10] suggesting that MAP3K6 may have a tumor suppressive role in cancers of the gastrointestinal tract. Our results demonstrate that inherited mutations in MAP3K6 may predispose individuals to gastric cancer; however, regulation of this pathway could have a broader role in both sporadic gastric cancers and carcinogenesis in general that warrants further investigation.

The penetrance of the disease in individuals carrying the p.P946L mutation is incomplete, with the mutation identified in 5 out of the 27 as yet unaffected individuals tested. Surprisingly, this included one individual who was homozygous for the mutation and was over the age of 80 with no reported cancer. However, there is significant variability in the age of onset (51-82 at diagnosis) in this family, unlike many other hereditary cancer syndromes, which feature an early age of onset. In particular, the average age of onset in this family is significantly higher than that typically diagnostic of HDGC. The Maritime Canadian p.P946L mutation may be hypomorphic, resulting in reduced function but not a complete loss of activity, contributing to the late age of onset and incomplete penetrance within the family, and offering a possible explanation for the individual who is homozygous for the mutation, but has not yet demonstrated signs of gastric cancer at the age of 80.

MAP3K6 Mutations in Gastric Cancer

The complexity between genotype and phenotype is illustrated in Familial Adenomatous Polyposis 2 (FAP2, MUTYH-associated Polyposis), where even in recessive cancer-predisposing syndromes the penetrance and age of onset can vary significantly between pathogenic mutations within the same gene. FAP2 is a recessive disorder caused by compound heterozygous or homozygous mutations in the mismatch repair gene mutY homolog (MUTYH). FAP2 is characterized by an extremely elevated risk of colorectal cancer (CRC). Average age of onset of CRC in FAP2 is 48-56. Penetrance for CRC among biallelic carriers has been estimated at approximately 80% by age 80 [26] due to the presence of homozygous carriers of of the common pathogenic p.T165C (p.T179C, rs34612342) mutation in the control group of at least two case-control studies [26,27]. This mutation is known to be pathogenic both in the homozygous state as well as a compound heterozygous mutation with other pathogenic variants. A homozygous individual for the other common FAP2 mutation, p.G396D (rs36053993), is also found in the Exome Sequencing Project (European-American) dataset. Interestingly, among CRC studies of MUTYH mutations while the p.T165C homozygous mutation is associated with a lower odds ratio it is generally also associated with a lower mean age of onset (48.9 versus 56.7) and more severe phenotype [28].

Incomplete penetrance of other recessive Mendelian disease is known in other cases including Leber Congenital Amaurosis [29], Schimke immuno-osseous dysplasia [30], and Bardet-Biedl syndrome [31,32]. In those cases it is expected that the homozygous carrier of a hypomorphic allele retains more protein function than more severe combinations of compound heterozygous or the hypomorphic allele in combination with complete loss of the other allele or its expression. In addition to hypomorphism, the homozygous combination of p.P946L alleles may also be displaying over-dominance as during stress signaling MAP3K6molecules likely form homo-dimers as well as heteromeric complexes with MAP3K5 [14,33]. If the p.P946L variant impacts this self-interaction the homozygous state may confer some protection over the heterozygous state.

Alternatively, the mutation may confer risk, but as with other cancers, additional environmental and genetic factors likely play a role in the progression to gastric cancer. This is consistent with experiments in MAP3K6 knock-out mice where cancer was only observed after the administration of both a carcinogen and an inflammatory agent, suggesting that the presence of additional stimuli are required [10]. Other variants segregating within the family may modulate both overall and individual risk of gastric cancer.

We do not know the importance of *H. pylori* in driving gastric cancer in the setting of this gene, as H. pylori infection was not a consistent finding among affected individuals from the Maritime Canadian family, and the rate was not different from what would be expected in the general population in this region [34,35]. It is possible that a second, as yet unidentified genetic abnormality not detected in the exome sequencing data is also contributing to disease in that pedigree. Several regions of the genome, including in the pedigree-wide analysis, displayed significant LOD or NPL scores. The region on chromosome 1, where MAP3K6 is located, was only identified in the reduced pedigree, where 2447 and 1845 were treated as unknown. Several regions identified pedigree-wide with high LOD scores (Table 1) are still of potential interest, and may indicate a second allele also segregating within the family contributing to the disease, with MAP3K6 mutations modifying the phenotype in the Maritime pedigree. Regions on chromosomes 3, 7, 17, and 20 displayed consistently high LOD scores in the pedigree-wide analysis, regardless of the specified penetrance (99% or 50%) while one on chromosome 16 was identified under both models but whose LOD score was more strongly affected by the penetrance parameter.

The truncating SNV identified in the Portuguese pedigree is much more likely to result in complete loss of function of MAP3K6, as the frameshift and truncation occur in the region of the protein involved in activation and interaction with MAP3K5. While penetrance of this mutation and segregation within the pedigree is unknown, penetrance is expected to be higher than that for the p.P946L mutation. This individual was diagnosed with Hereditary Diffuse Gastric Cancer at the age of 62, intermediate between early-onset HDGC and the late-onset observed in the Maritime Canadian pedigree. The impact of the other germline variants discovered in the additional probands is not clear.

The identification of new germline mutations that appear to predispose individuals to familial gastric cancer can aid in identifying at risk individuals in affected families that are negative for *CDH1* mutations. It may also help to shed light on the underlying mechanisms leading to cancer development. Further, somatic mutations and altered expression of *MAP3K6* in sporadic cancers and gastric cancer cell lines may suggest *MAP3K6* as a potential therapeutic target for exploration. Its binding partner in the heterometric complex, *MAP3K5*, is already being investigated for therapeutic potential [36–39].

The presence of MAP3K6 mutations in the probands of six unrelated families, somatic mutations in sporadic cancers (particularly those of the gastrointestinal tract), evidence from MAP3K6knockout mice, second-hit mutations or hypermethylation of the wild-type allele in the tumors tested, and its molecular role in inflammation and apoptosis, all suggest that MAP3K6 is an interesting candidate for mutations associated with Familial Gastric Cancer, warranting further study in additional cohorts of CDH1 mutation-negative familial gastric cancer cases.

Methods

Ethics Statement

Approval for the research study was obtained from the IWK Health Centre research ethics board (project approval number 1005367). Informed consent was obtained from individuals or their guardians for all samples used in this study. DNA was obtained from blood, saliva or FFPE samples using standard methods

Data Availability

Due to the small number of samples and nature of the study, genetic information can reveal potentially identifiable and unrelated health data of individuals from the family, including individuals who were not enrolled in this study. For this reason, the research ethics approval of this study and informed consent signed by participants does not allow for data to be deposited in public databases. Data used in this study are available upon request from the corresponding author pending approval from the Maritime Medical Genetics Service at the IWK Health Science Centre at: Maritime Medical Genetics Service, PO Box 9700, Halifax, Nova Scotia, Canada, B3K 6R8. Phone +1-902-470-8754.

SNP Genotyping and Genomic Mapping

Whole genome high-density single nucleotide polymorphism (SNP) genotype scanning was performed at the McGill University and Genome Quebec Centre for Innovation, using the Illumina Human610-Quadv1_B chip (Illumina, Inc., San Diego, CA) panel with 620,901 markers and the Illumina HumanOmni 2.5M panel. DNA for SNP genotyping was isolated from either saliva or blood

collected from patients. Genotype arrays were scanned using the Bead Array Reader (Illumina, Inc.), plate Crane Ex, and Illumina BeadLab software (Illumina, Inc.). Initial quality control and export of data was done using Illumina's GenomeStudio software.

Before linkage analysis was performed, the set of SNPs was pruned to obtain a subset that was appropriate for linkage analysis. Out of the 2,391,739 markers passing QC, only markers on the autosomes were retained. Markers with alleles ambiguous for strand information (A/T and G/C variants) were removed, in order to facilitate strand matching with HapMap data. Markers that were monomorphic in all genotyped samples were removed. At this point, 1,008,604 markers remained. This set of SNPs was merged with HapMap3 CEU data; only markers present in both sets of data were kept, matching on marker name. This resulted in 476,551 markers. From this set, only markers with MAF>0.4 were kept. In order to avoid inconsistencies between the observed data and the genetic map, markers with unique positions on the genetic map were selected. Arbitrarily, for a group of markers with the same genetic position, the marker with the lowest physical position was retained, and the remaining markers in the group were discarded. Finally, since linkage disequilibrium (LD) between the SNPs can arbitrarily inflate multipoint LOD scores [40], the markers were pruned to remove strong pairwise LD ($r^2 < 0.1$ on a chromosome). This resulted in a set of 8,472 SNPs across the genome that were roughly independent, and should have high informativity for linkage analysis. For this set, the average intermarker distance was 0.57 cM, or 451 kb. Filtering was performed using a combination of PLINK v1.07 [41] and inhouse scripts.

Merlin 1.1.2 [12] was used to perform multipoint nonparametric linkage on the family, using the set of 8,472 SNPs selected above. Only affected individuals will contribute to this analysis; the individuals with unknown affection status will only help infer phase. The S_{all} statistic was used, which looks for allele sharing in all affected individuals, and tends to perform well for dominant traits [42]. The exponential model of Kong and Cox was used, which is preferred when a small number of families are analyzed [43]. CEU genotypes from HapMap3 release 28 [44] were used to estimate allele frequencies. Since it is possible that sample 2447, a distant cousin to the other affected individuals, had a slightly different phenotype, the analysis was repeated, coding this individual as having an unknown affection status, effectively removing him from the analysis. For the pedigree-wide analysis regions with an NPL score >2 were selected, with boundaries defined by markers flanking a set of SNPs with p < 0.05. For the reduced pedigree with sample 2447 excluded regions with an NPL score >1 were selected.

Parametric linkage analysis was also performed, under two dominant models, one with 50% and the other with 99% penetrance. Both models used a disease allele frequency of 0.001 and 2% phenocopy rate. The analyses were performed using Merlin. Regions with LOD>1 and with SNP boundaries defined by 1 – LOD support interval were used for further analysis. Similarly to the NPL analysis above, two sets of analyses were performed with 2447 coded as either affected or unknown.

Whole Exome Sequencing

A total of 3 μ g of DNA was used for exome capture with the Agilent SureSelect All Exon 38 Mb kit. Sequencing was performed with 100-bp paired-end reads using the Illumina HiSeq 2000 at the McGill University and Genome Quebec Centre for Innovation as previously described [45]. Reads were assembled against the human genomic reference sequence (hg19) using the Burrows-Wheeler Aligner (BWA) [46]. Genomic variants were

called using the Genome Analysis Toolkit (GATK) pipeline [47], and annotated with SnpEff [48] and GEMINI [49]. All variants were compared against dbSNP [50], 1000 Genomes Project [51], the Exome Sequencing Project (Exome Variant Server, NHLBI GO Exome Sequencing Project (ESP): http://evs.gs.washington. edu/EVS [Accessed October, 2013]), and a pool of exomes sequenced at the Genome Quebec Innovation Centre (1532 samples). Potentially damaging variants included non-synonymous mutations (missense and nonsense), splice-site variants, and frameshift changes due to insertions and/or deletions (indels). Exome variants were further filtered by their location in regions of the genome not excluded by linkage analysis. Variants were further selected based on their predicted impact by snpEff and GEMINI (Medium and High Impact) as well as their minor allele frequency in public databases and additional exomes sequenced at Genome Quebec (MAF $\leq 2\%$). Selected mutations in candidate genes were verified and screened in additional family members by Sanger sequencing.

PCR and Direct Sanger Sequencing

Selected regions were amplified from genomic DNA by PCR. Amplified fragments were purified from an agarose gel, and sequenced using Sanger fluorescent sequencing and capillary electrophoresis. Sequence traces were analyzed using Mutation-Surveyor V.3.97 (Soft Genetics, Inc.). Because of a lack of a fresh DNA sample, initially the three children of 1884 and his spouse, 1885 were used for genotyping. The presence of the heterozygous mutation (c.[2837C>T]; [=]) at the genomic level could be inferred for the affected individual 1884 by the absence of the mutation in his spouse, 1885, but the presence of the mutation in their child, 1903 and not in their children 1902 and 1879. The mutation was then confirmed in DNA derived from a tumorous section of an FFPE sample from this individual. An additional second-hit mutation was observed in this FFPE-derived DNA, which could be inferred to be somatic based on its absence in all three children despite both paternal copies being represented among them. In the Portuguese family, FFPE-derived DNA was extracted from the family proband's tumor (macrodissected to guarantee a minimum of 75% of tumor cells) and was used for somatic mutation analysis of the full MAP3K6 coding sequence and intron-exon boundaries. LOH was inferred from the Sanger sequencing data, using the germline mutation site as an intragenic marker. LOH would have been considered if only the mutant allele had been found in the proband's tumor.

Validation in an Unrelated Population

DNA previously isolated from the blood of 115 FGC patients that had been demonstrated to be negative for CDH1 mutations were sequenced for the presence of mutations in MAP3K6 using a multiplexed TruSeq Custom Amplicon (TSCA) sequencing assay (Illumina, San Diego). Samples were first quantified using Qubit dsDNA broad ranged assay kit (Life Technologies) and custom oligos were pooled and hybridized to individual samples. Sample indices were then added to each template library by PCR using the TSCA reagents and protocol (Illumina, San Diego). A post-PCR bead-based normalization technique was used according to the TSCA protocol to avoid the necessity of laborious and timeconsuming quantification methods. Samples were then pooled prior to loading onto the MiSeq platform for simultaneous cluster generation, bi-directional sequencing and data analysis. Amplicons were designed only for coding regions (exons) of the targeted areas of interest plus up to 25 base pairs of padding around each individual amplicon. Coverage across each amplicon varied depending on location and quality of sample. In the sample

where the germline MAP3K6 frameshift variant was found, there was $80 \times$ coverage of the amplicon of interest.

Detection of MAP3K6 Hypermethylation

We have submitted the MAP3K6 genomic sequence to a CpG island searcher (http://cpgislands.usc.edu/, [18]) and identified two CpG islands that were also annotated in ENSEMBL (release 75; www.ensembl.org). CpG island 1 was located at the promoter region and CpG island 2 encompassed exon 10 and part of the downstream intron. MAP3K6 methylation analysis was performed in the Portuguese family proband, for which FFPE tumor and peripheral blood lymphocytes (PBLs) DNA was available. Additionally, we analyzed DNA from normal stomach mucosa from controls (n = 4), and 7 gastric cancer cell lines DNA (MKN28, MKN45, NCI-N87, Kato III, AGS, GP202 and IPA220). The EpiTect Bisulfite Kit (Qiagen, Valencia, Calif) was used to treat 300 ng of DNA per sample. Unmethylated cytosines were converted to uracil, whereas methylated ones remained unmodified. A fraction of CpG sites contained within CpG islands 1 and 2 were PCR amplified using flanking primers, specifically designed for bisulfite treated DNA sequences without CpG sites, and sequenced for methylation status determination (primer sequences available upon request). Independent PCR reactions were performed at least twice for each sample.

Predicting Pathogenicity of Detected Mutations

For all mutations detected in this work, either germline or somatic, the potential pathogenicity was predicted using a variety of published bioinformatic methods. Because individual tools are known to have differing performance profiles (both in terms of false positives and false negatives), we employed a variety of different tools that use differing algorithms for our assessment. In this work we evaluated all relevant mutations using PMut [52], PolyPhen2 [20,53], SIFT [21], Provean [54], MutationTaster [55], EvoD [19], and FATHMM [56]. In particular FATHMM has been updated [57] with an algorithm specific to somatic mutations in cancer, which has also been used here. Unless otherwise noted default options were used with all prediction programs for their default web interface. Both PROVEAN and SIFT scores were provided through the PROVEAN interface.

References

- Ferlay J, Shin H-R, Bray F, Forman D, Mathers C, et al. (2010) Estimates of worldwide burden of cancer in 2008: GLOBOCAN 2008. 127: 2893–2917. doi:10.1002/ijc.25516.
- Wadhwa R, Song S, Lee J-S, Yao Y, Wei Q, et al. (2013) Gastric cancermolecular and clinical dimensions. Nat Rev Clin Oncol 10: 643–655. doi:10.1038/nrclinonc.2013.170.
- Lauren P (1965) The two histological main types of gastric carcinoma, an attempt at a histoclinical classification. Acta Pathol Microbiol Scand 64: 31– 40.
- Parkin DM (2006) The global health burden of infection-associated cancers in the year 2002. Int J Cancer 118: 3030–3044. doi:10.1002/ijc.21731.
- Oliveira C, Seruca R, Carneiro F (2009) Hereditary gastric cancer. 23: 147–157. doi:10.1016/j.bpg.2009.02.003.
- Guilford P, Hopkins J, Harraway J, McLeod M, McLeod N, et al. (1998) Ecadherin germline mutations in familial gastric cancer. Nature 392: 402–405. doi:10.1038/32918.
- Fitzgerald RC, Hardwick R, Huntsman D, Carneiro F, Guilford P, et al. (2010) Hereditary diffuse gastric cancer: updated consensus guidelines for clinical management and directions for future research. J Med Genet 47: 436–444. doi:10.1136/jmg.2009.074237.
- Park J-G, Yang H-K, Kim WH, Caldas C, Yokota J, et al. (2000) Report on the first meeting of the International Collaborative Group on Hereditary Gastric Cancer. J Natl Cancer Inst 92: 1781–1782. doi:10.1093/jnci/92.21.1781.
- Oliveira C, Senz J, Kaurah P, Pinheiro H, Sanges R, et al. (2009) Germline CDH1 deletions in hereditary diffuse gastric cancer families. Hum Mol Genet 18: 1545–1555. doi:10.1093/hmg/ddp046.

Supporting Information

Figure S1 Partial representative electropherograms from the methylation analysis of MAP3K6 CpG island 2. (PDF)

Figure S2 Partial electropherogram showing the truncating mutation (2544delC) and the wild-type sequence at similar detection levels in the tumour from the Portuguese proband. This result allows exclusion of LOH as a second-hit, due to maintained heterozygosity at the mutation site. (PDF)

Table S1 Summary of variants identified in the exome sequences of 1826, 1841, and 2447, located in the regions identified by parametric linkage in the entire pedigree and that met the filtering criteria described in the Methods. Variants were then sanger sequenced in other affected individuals. (XLSX)

Table S2 Summary of variants identified in the exome sequences of 1826 and 1841 that were located in the regions identified by non-parametric linkage in the sub-pedigree represented by the proband, proband's mother, and proband's maternal aunts. Variants were filtered as described in the Methods and select variants were sanger sequenced in other affected individuals within the sub-pedigree. (XLSX)

Acknowledgments

We are grateful to the family members who generously contributed their time and materials for this research. The authors would like to acknowledge the significant contributions to this work that were made by Dr. Duane Guernsey. He was a valued member of our research team who passed away during the course of this project.

Author Contributions

Conceived and designed the experiments: DG KB SH DH CO AO JM ML MES CRM CVF. Performed the experiments: DG SH MN HJ HP CO WYH WG PK. Analyzed the data: DG SH KB MN DH JM CO HP MAL SW. Contributed reagents/materials/analysis tools: KB CM DH CO JM LSP ML MES WYH PS ALR GS AO SD. Wrote the paper: DG KB WYH SH DH ALR PS CRM CO CVF AO MES. Coordinated clinical ascertainment: CM ALR PS SD SB LSP ML AO.

- Iriyama T, Takeda K, Nakamura H, Morimoto Y, Kuroiwa T, et al. (2009) ASK1 and ASK2 differentially regulate the counteracting roles of apoptosis and inflammation in tumorigenesis. The EMBO Journal 28: 843–853. doi:10.1038/ emboj.2009.32.
- Zang ZJ, Ong CK, Cutcutache I, Yu W, Zhang SL, et al. (2011) Genetic and structural variation in the gastric cancer kinome revealed through targeted deep sequencing. Cancer Res 71: 29–39. doi:10.1158/0008-5472.CAN-10-1749.
- Abecasis GR, Cherny SS, Cookson WO, Cardon LR (2002) Merlin–rapid analysis of dense genetic maps using sparse gene flow trees. Nature Genetics 30: 97–101. doi:10.1038/ng786.
- Forbes SA, Bindal N, Bamford S, Cole C, Kok CY, et al. (2011) COSMIC: mining complete cancer genomes in the Catalogue of Somatic Mutations in Cancer. Nucleic Acids Res 39: D945–D950. doi:10.1093/nar/gkq929.
- Takeda K, Shimozono R, Noguchi T, Umeda T, Morimoto Y, et al. (2007) Apoptosis signal-regulating kinase (ASK) 2 functions as a mitogen-activated protein kinase kinase kinase in a heteromeric complex with ASK1. Journal of Biological Chemistry 282: 7522–7531. doi:10.1074/jbc.M607177200.
- Eto N, Miyagishi M, Inagi R, Fujita T, Nangaku M (2009) Mitogen-activated protein 3 kinase 6 mediates angiogenic and tumorigenic effects via vascular endothelial growth factor expression. Am J Pathol 174: 1553–1563. doi:10.2353/ajpath.2009.080190.
- Baylin SB, Esceller M, Rountree MR, Bachman KE, Schuebel K, et al. (2001) Aberrant patterns of DNA methylation, chromatin formation and gene expression in cancer. Hum Mol Genet 10: 687–692. doi:10.1093/hmg/ 10.7.687.

- Choi MR, In Y-H, Park J, Park T, Jung KH, et al. (2012) Genome-scale DNA methylation pattern profiling of human bone marrow mesenchymal stem cells in long-term culture. Exp Mol Med 44: 503. doi:10.3858/emm.2012.44.8.057.
- Takai D, Jones PA (2003) The CpG island searcher: a new WWW resource. In silico biology 3: 235–240.
- Kumar S, Sanderford M, Gray VE, Ye J, Liu L (2012) Evolutionary diagnosis method for variants in personal exomes. Nat Methods 9: 855–856. doi:10.1038/ nmeth.2147.
- Adzhubei IA, Schmidt S, Peshkin L, Ramensky VE, Gerasimova A, et al. (2010) A method and server for predicting damaging missense mutations. Nat Methods 7: 248–249. doi:10.1038/nmeth0410-248.
- Ng PC, Henikoff S (2003) SIFT: Predicting amino acid changes that affect protein function. Nucleic Acids Res 31: 3812–3814. doi:10.1093/nar/gkg509.
- Kulis M, Queirós AC, Beekman R, Martín-Subero JI (2013) Intragenic DNA methylation in transcriptional regulation, normal differentiation and cancer. Biochimica et Biophysica Acta (BBA) - Gene Regulatory Mechanisms 1829: 1161–1174. doi:10.1016/j.bbagrm.2013.08.001.
- Hayakawa Y, Hirata Y, Nakagawa H (2011) Apoptosis signal-regulating kinase 1 and cyclin D1 compose a positive feedback loop contributing to tumor growth in gastric cancer Vol. 108. pp. 780–785. doi:10.1073/pnas.1011418108/-/DC Supplemental.
- Greenman C, Stephens P, Smith R, Dalgliesh GL, Hunter C, et al. (2007) Patterns of somatic mutation in human cancer genomes. Nature 446: 153–158. doi:10.1038/nature05610.
- Sjöblom T, Jones S, Wood LD, Parsons DW, Lin J, et al. (2006) The consensus coding sequences of human breast and colorectal cancers. Science 314: 268–274. doi:10.1126/science.1133427.
- Lubbe SJ, Di Bernardo MC, Chandler IP, Houlston RS (2009) Clinical implications of the colorectal cancer risk associated with MUTYH mutation. Journal of Clinical Oncology 27: 3975–3980. doi:10.1200/JCO.2008.21.6853.
- Cleary SP, Cotterchio M, Jenkins MA, Kim H, Bristow R, et al. (2009) Germline MutY Human Homologue Mutations and Colorectal Cancer: A Multisite Case-Control Study. Gastroenterology 136: 1251–1260. doi:10.1053/j.gastro.2008. 12.050.
- Theodoratou E, Campbell H, Tenesa A, Houlston R, Webb E, et al. (2010) A large-scale meta-analysis to refine colorectal cancer risk estimates associated with MUTYH variants. Br J Cancer 103: 1875–1884. doi:10.1038/sj.bjc.6605966.
- Siemiatkowska AM, Schuurs-Hoeijmakers JHM, Bosch DGM, Boonstra FN, Riemslag FCC, et al. (2014) Nonpenetrance of the Most Frequent Autosomal Recessive Leber Congenital Amaurosis Mutation in NMNAT1. JAMA Ophthalmol 132: 1002–4. doi:10.1001/jamaophthalmol.2014.983.
- Baradaran-Heravi A, Cho KS, Tolhuis B, Sanyal M, Morozova O, et al. (2012) Penetrance of biallelic SMARCAL1 mutations is associated with environmental and genetic disturbances of gene expression. Hum Mol Genet 21: 2572–2587. doi:10.1093/hmg/dds083.
- Estrada-Cuzcano A, Koenekoop RK, Senechal A, De Baere EBW, de Ravel T, et al. (2012) BBS1 Mutations in a Wide Spectrum of Phenotypes Ranging From Nonsyndromic Retinitis Pigmentosa to Bardet-Biedl Syndrome. Arch Ophthalmol 130: 1425–1432. doi:10.1001/archophthalmol.2012.2434.
- Beales PL, Badano JL, Ross AJ, Ansley SJ (2003) Genetic Interaction of BBS1 Mutations with Alleles at Other BBS Loci Can Result in Non-Mendelian Bardet-Biedl Syndrome. Am J Hum Gen. doi:10.1086/375178.
- Ortner E, Moelling K (2007) Heteromeric complex formation of ASK2 and ASK1 regulates stress-induced signaling. Biochemical and Biophysical Research Communications 362: 454–459. doi:10.1016/j.bbrc.2007.08.006.
- van Zanten SJOV, Pollak PT, Best LM, Bezanson GS, Marrie T (1994) Increasing Prevalence of Helicobacter pylori Infection with Age: Continuous Risk of Infection in Adults Rather than Cohort Effect. J Infect Dis 169: 434–437. doi:10.1093/infdis/169.2.434.
- Goodman KJ, Jacobson K, Veldhuyzen van Zanten S (2008) Helicobacter pylori infection in Canadian and related Arctic Aboriginal populations. Can J Gastroenterol 22: 289–295.
- Singh O, Shillings A, Craggs P, Wall I, Rowland P, et al. (2013) Crystal structures of ASK1-inhibtor complexes provide a platform for structure-based drug design. Protein Science 22: 1071–1077. doi:10.1002/pro.2298.

- Jin L, Li D, Lee JS, Elf S, Alesi GN, et al. (2013) p90 RSK2 Mediates Antianoikis Signals by both Transcription-Dependent and-Independent Mechanisms. ... and cellular biology. doi:10.1128/MCB.01677-12.
- Cai D-T, Jin H, Xiong Q-X, Liu W-G, Gao Z-G, et al. (2013) ER stress and ASK1-JNK activation contribute to oridonin-induced apoptosis and growth inhibition in cultured human hepatoblastoma HuH-6 cells. Mol Cell Biochem 379: 161–169. doi:10.1007/s11010-013-1638-2.
- Hayakawa R, Hayakawa T, Takeda K (2012) Therapeutic targets in the ASK1dependent stress signaling pathways Vol. 88. pp. 434–453. doi:10.2183/pjab. 88.434.
- Huang Q, Shete S, Amos CI (2004) Ignoring linkage disequilibrium among tightly linked markers induces false-positive evidence of linkage for affected sib pair analysis. Am J Hum Gen 75: 1106–1112. doi:10.1086/426000.
- Purcell S, Neale B, Todd-Brown K, Thomas L, Ferreira MAR, et al. (2007) PLINK: a tool set for whole-genome association and population-based linkage analyses. Am J Hum Gen 81: 559–575. doi:10.1086/519795.
- Whittemore AS, Halpern J (1994) A class of tests for linkage using affected pedigree members. Biometrics 50: 118–127.
- Kong A, Cox NJ (1997) Allele-sharing models: LOD scores and accurate linkage tests. Am J Hum Gen 61: 1179–1188. doi:10.1086/301592.
- International HapMap 3 Consortium, Altshuler DM, Gibbs RA, Peltonen L, Dermitzakis E, et al. (2010) Integrating common and rare genetic variation in diverse human populations. Nature 467: 52–58. doi:10.1038/nature09298.
- Alfares A, Nunez LD, Al-Thihli K, Mitchell J, Melançon S, et al. (2011) Combined malonic and methylmalonic aciduria: exome sequencing reveals mutations in the ACSF3 gene in patients with a non-classic phenotype. 48: 602– 605. doi:10.1136/jmedgenet-2011-100230.
- Li H, Durbin R (2009) Fast and accurate short read alignment with Burrows-Wheeler transform. Bioinformatics 25: 1754–1760. doi:10.1093/bioinformatics/ btp324.
- McKenna A, Hanna M, Banks E, Sivachenko A, Cibulskis K, et al. (2010) The Genome Analysis Toolkit: a MapReduce framework for analyzing nextgeneration DNA sequencing data. Genome Research 20: 1297–1303. doi:10.1101/gr.107524.110.
- Cingolani P, Platts A, Wang LL, Coon M, Nguyen T, et al. (2012) A program for annotating and predicting the effects of single nucleotide polymorphisms, SnpEff: SNPs in the genome of Drosophila melanogaster strain w1118; iso-2; iso-3. Fly (Austin) 6: 80–92. doi:10.4161/fly.19695.
- Paila U, Kirchner R, Chapman BA, Quinlan AR (2013) GEMINI: integrative exploration of genetic variation and genome annotations. PLoS Comput Biol 9: e1003153. doi:10.1371/journal.pcbi.1003153.
- Sherry ST, Ward MH, Kholodov M, Baker J, Phan L, et al. (2001) dbSNP: the NCBI database of genetic variation. Nucleic Acids Res 29: 308–311. doi: 10.1093/nar/29.1.308.
- The 1000 Genomes Project Consortium (2010) A map of human genome variation from population-scale sequencing. Nature 467: 1061–1073. doi:10.1038/nature09534.
- Ferrer-Costa C, Gelpí JL, Zamakola L, Parraga I, la Cruz de X, et al. (2005) PMUT: a web-based tool for the annotation of pathological mutations on proteins. Bioinformatics 21: 3176–3178. doi:10.1093/bioinformatics/bti486.
- Adzhubei I, Jordan DM, Sunyaev SR (2013) Predicting functional effect of human missense mutations using PolyPhen-2. Curr Protoc Hum Genet Chapter 7: Unit7.20–7.20.41. doi:10.1002/0471142905.hg0720s76.
- Choi Y, Sims GE, Murphy S, Miller JR, Chan AP (2012) Predicting the functional effect of amino acid substitutions and indels. 7: e46688. doi:10.1371/ journal.pone.0046688.
- Schwarz JM, Rödelsperger C, Schuelke M, Seelow D (2010) MutationTaster evaluates disease-causing potential of sequence alterations. Nat Methods 7: 575– 576. doi:10.1038/nmeth0810-575.
- Shihab HA, Gough J, Cooper DN, Stenson PD, Barker GLA, et al. (2013) Predicting the Functional, Molecular, and Phenotypic Consequences of Amino Acid Substitutions using Hidden Markov Models. Hum Mutat 34: 57–65. doi:10.1002/humu.22225.
- Shihab HA, Gough J, Cooper DN, Day INM, Gaunt TR (2013) Predicting the functional consequences of cancer-associated amino acid substitutions. Bioinformatics 29: 1504–1510. doi:10.1093/bioinformatics/btt182.